Author's response to reviews

Title: Risk of prostate cancer after detection of isolated high-grade prostatic intraepithelial neoplasia (HGPIN) on extended core needle biopsy: a UK hospital experience

Authors:
- Paras B Singh (parassingh@btinternet.com)
- Caroline M Nicholson (caroline.nicholson@lthtr.nhs.uk)
- Narasimhan Ragavan (n_ragavan@yahoo.com)
- Rosemary A Blades (rosie.blades@lthtr.nhs.uk)
- Francis L Martin (f.martin@lancaster.ac.uk)
- Shyam S Matanhelia (shyam.matanhelia@lthtr.nhs.uk)

Version: 5 Date: 5 May 2009

Author's response to reviews: see over
Dear Sir,

Thank you for considering our manuscript entitled “Risk of prostate cancer after isolated high-grade prostatic intraepithelial neoplasia (HGPIN) detected on extended core needle biopsy: a UK hospital experience” (MS: 1761119029239237) by Singh et al. for publication in BMC Urology. We have attached our manuscript with the following suggested formatting changes (suggested formatting changes are highlighted in red):

**Major revisions (we require the author to make these changes).**

Title - please ensure that the title is identical in the manuscript and the submission system
We can confirm this.

Acknowledgements - please add an Acknowledgements section after the Authors' contributions and before the References, and move the statement on financial support to the Acknowledgements.
We have made these changes in the re-submitted manuscript.

Table vertical elements - we are unable to display vertical lines within the tables. Please re-layout your table 2 without the vertical lines.
We have revised the table layout.

**Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).**

Remove the Keywords from the title page.
We have addressed this in the submitted manuscript.

Patients & Methods - Please rename this section 'Methods' in both the abstract and the main text of the manuscript.
Again this has been revised.

Abstract - Please ensure that the Abstract in the manuscript matches exactly what is in the submission system.
We have ensured this.

Highlighting/tracking - Please remove all highlighting and tracking from the manuscript.
Again we have addressed this in the submitted manuscript.

Ethical approval - please move this to the Methods section.
This has been addressed.
Authors' contribution - please change this section heading to Authors' contributions. We have made the changes.

Legend to Figure - please change this section heading to Figure legends. We have made these changes.

Once again thank you for handling our manuscript and giving us the opportunity to make the changes.

Yours sincerely,

Paras B Singh